NOVARTIS AG Form 6-K May 02, 2012

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 6-K

REPORT OF FOREIGN PRIVATE ISSUER PURSUANT TO RULE 13a-16 or 15d-16 OF THE SECURITIES EXCHANGE ACT OF 1934

Report on Form 6-K dated May 1, 2012 (Commission File No. 1-15024)

Novartis AG

(Name of Registrant)

Lichtstrasse 35

4056 Basel

Switzerland

(Address of Principal Executive Offices)

Indicate by check mark whether the registrant files or will file annual reports under cover of Form 20-F or Form 40-F:

Form 20-F: x	Form 40-F: o
Indicate by check mark if the registrant is submitting the Form 6-K in pa	aper as permitted by Regulation S-T Rule 101(b)(1):
Yes: o	No: x
Indicate by check mark if the registrant is submitting the Form 6-K in pa	aper as permitted by Regulation S-T Rule 101(b)(7):
Yes: o	No: x
Indicate by check mark whether the registrant by furnishing the informate the Commission pursuant to Rule 12g3-2(b) under the Securities Exchange.	
Yes: o	No: x

Novartis International AG Novartis Global Communications CH-4002 Basel Switzerland

http://www.novartis.com

MEDIA RELEASE • COMMUNIQUE AUX MEDIAS • MEDIENMITTEILUNG

• Lucentis (¡	Two year data confirms an overall significantly higher risk of serious systemic adverse events with unlicensed bevacizumab versus p=0.004)
• vascular ā	Reports of arteriothrombotic events, systemic hemorrhage, congestive heart failure, venous thrombotic events, hypertension and leath were more frequent in bevacizumab treated patients (p =0.07)
• compared	As in year one, there were significantly more gastrointestinal disorders in patients treated with unlicensed intravitreal bevacizumab to Lucentis $(p=0.005)$
•	More than one million patient treatment years globally inform the well characterized safety profile of Lucentis since its 2006 launch

Basel, Mai 1, 2012 Recently announced results from two-year CATT(1) (Comparison of Age-related macular degeneration Treatment Trials) adds to the growing body of evidence, including CATT 1-year data and two retrospective database analyses(2),(3),(4), suggesting serious safety concerns associated with the use of unlicensed intravitreal Avastin® (bevacizumab).

The full two year data from this study of 1107 patients with wet AMD shows a 30% higher risk in overall serious systemic adverse events with unlicensed intravitreal bevacizumab versus Lucentis (p=0.004). This is a similar finding to the year one data(2). Furthermore, the study highlights reports of arteriothrombotic events, systemic hemorrhage, congestive heart failure, venous thrombotic events, hypertension and vascular death that were more frequent in bevacizumab treated patients (p=0.07). These events have been previously associated with systemic anti-VEGF treatment.

In addition, as in year one, there were significantly more gastrointestinal disorders in patients treated with unlicensed intravitreal bevacizumab compared to Lucentis (p=0.005). As the CATT authors comment, this has been an area of concern in previous studies of systemic bevacizumab(1).

This data adds to the growing body of evidence suggesting that the overall risk of serious ocular and systemic side effects is higher with unlicensed intravitreal bevacizumab compared to Lucentis. However, the studies were not powered to assess differences in infrequent but serious events such as death and stroke, which were found in previously published large Medicare database analyses to be significantly more frequent with unlicensed Avastin, said Tim Wright, Global Head of Development, Novartis Pharma. The apparent differential safety risk between the two medicines may be due to differences in the molecules and their commercial formulation. This underscores the importance of drug design and development with the patient and disease process in mind.

Previously published retrospective Medicare analyses compared the safety of unlicensed ocular bevacizumab and Lucentis for treating wet AMD(3),(4). The primary analysis from the Curtis study showed that unlicensed bevacizumab and Lucentis were not associated with increased risks of death, heart attack, bleeding or stroke compared with photodynamic therapy or pegaptanib. A secondary analysis in the Curtis et al study, which included approximately 40,000 patients, showed a significantly higher risk of mortality (16%) and stroke (28%) with unlicensed intravitreal bevacizumab than with Lucentis. Gower et al. assessed 77,886 Medicare beneficiaries with wet AMD over 10 months. In this study patients treated with unlicensed intravitreal bevacizumab experienced an 11% significantly increased risk of death and a 57% significantly increased risk of hemorrhagic stroke compared to patients treated with Lucentis. The risk of ocular inflammation was 80% higher with unlicensed intravitreal bevacizumab than with Lucentis.

In addition to the serious systemic safety concerns, in the past year since the first year CATT results were presented, there have been reports of additional clusters of severe eye inflammation, some resulting in blindness, following unlicensed intravitreal bevacizumab use across the United States and in the United Kingdom(5),(6).

These important findings and continued reported clusters of eye inflammation have been recognized by several health authorities and other organizations around the world, including Sweden s Medical Products Agency and Health Canada, resulting in increased safety warnings on the use of unlicensed intravitreal bevacizumab(7),(8),(9).

Irrespective of the results of CATT or any other head-to-head study, bevacizumab remains unlicensed for any ocular condition, for administration in the eye and compounding into smaller doses.

Novartis puts patient safety first and believes Lucentis is the best treatment option for patients with wet AMD. Novartis believes that, within regulatory guidelines, the unlicensed use of drugs should be limited to cases where there is an unmet medical need which cannot be fulfilled by licensed medications.

More information on CATT two-year results(1)

At two years, patients receiving as needed Lucentis fully maintained first-year vision gains with an average 5.7 injections in the second year. In contrast, patients treated with as needed bevacizumab experienced a greater decline in vision despite receiving significantly more injections over the two year period (p=0.01). Secondary anatomical outcomes were significantly better with Lucentis compared to unlicensed intravitreal bevacizumab. In general, outcomes were directionally favorable for Lucentis compared to unlicensed intravitreal bevacizumab.

CATT (Comparison of Age-related macular degeneration Treatment Trials): Lucentis-Avastin Trial is a two-year, prospective, multicenter, randomized, non-inferiority clinical trial involving four treatment arms. The study, evaluating the relative efficacy and safety of Lucentis and Avastin for the treatment of wet AMD, was funded by the National Eye Institute (NEI) and was conducted in the United States. The primary endpoint was the mean change in visual acuity (VA) at month 12 compared to baseline as measured on a standard ETDRS (Early Treatment Diabetic Retinopathy Study) chart.

About Lucentis® (ranibizumab)

Lucentis is a humanized therapeutic antibody fragment designed to block all biologically active forms of vascular endothelial cell growth factor-A (VEGF-A). Increased levels of VEGF-A are seen in wet age-related macular degeneration (AMD) and other ocular diseases such as diabetic macular edema (DME) and retinal vein occlusion (RVO). Lucentis has been designed, developed and formulated specifically for use in ocular disease with the aim of stabilizing and improving visual acuity in these patients.

Lucentis is licensed for the treatment of wet AMD in more than 100 countries, and in more than 60 countries for the treatment of visual impairment due to DME, and visual impairment due to macular edema secondary to RVO, including both branch- and central-RVO. In many countries, including in Europe, Lucentis has an individualized treatment regimen with the goal of maximizing visual outcomes while minimizing under- or over-treating patients.

In 2011, Novartis launched the Luminous (TM) program, one of the largest observational studies in ophthalmology, aiming to recruit over 30,000 patients from clinics across Asia, Australia, Europe, North and South America to further broaden the understanding of ocular disease and the use of Lucentis in its approved indications. Luminous is a five-year observational, international, multicenter program that is expected to provide long-term effectiveness and safety data for Lucentis as well as assess the treatment patterns and health-related quality of life issues of patients treated with Lucentis.

Novartis works closely with reimbursement authorities to ensure that Lucentis is cost effective for appropriate patients and has an excellent track record of working with healthcare systems to secure access to medicines for patients. In fact, Lucentis has been shown to be cost effective and is reimbursed for the treatment of wet AMD in most countries, including a recommendation by the National Institute for Health and Clinical Excellence (NICE) in the UK.

Lucentis has a well-characterized safety profile and Novartis systematically monitors the safety and tolerability of Lucentis for licensed indications on an ongoing basis. Its safety profile has been well established in a clinical development program that enrolled more than 10,000 patients across indications.

Serious adverse events related to the injection procedure include endophthalmitis, retinal detachment, retinal tear and traumatic cataract. Other serious ocular events observed among Lucentis-treated patients included intraocular inflammation and increased intraocular pressure. Non-eye related serious side effects, although not common, include heart attacks, strokes and death.

Lucentis was developed by Genentech and Novartis. Genentech has the commercial rights to Lucentis in the United States. Novartis has exclusive rights in the rest of the world. Lucentis is a registered trademark of Genentech Inc.

Disclaimer

The foregoing release contains forward-looking statements that can be identified by terminology such as believes, launched, aiming, to further broaden, expected, or similar expressions, or by express or implied discussions regarding potential future revenues from Lucentisu should not place undue reliance on these statements. Such forward-looking statements reflect the current views of management regarding future events, and involve known and unknown risks, uncertainties and other factors that may cause actual results with Lucentis to be materially different from any future results, performance or achievements expressed or implied by such statements. There can be no guarantee that Lucentis will achieve any particular levels of revenue in the future. In particular, management s expectations regarding Lucentis could be affected by, among other things, unexpected clinical trial results, including unexpected new clinical data and unexpected additional analysis of existing clinical data, including with respect to the CATT study, and the Luminous program; unexpected regulatory actions or delays or government regulation generally; competition in general; government, industry and general public pricing pressures; the company s ability to obtain or maintain patent or other proprietary intellectual property protection; unexpected manufacturing issues; the impact that the foregoing factors could have on the values attributed to the Novartis Group's assets and liabilities as recorded in the Group's consolidated balance sheet, and other risks and factors

referred to in Novartis AG s current Form 20-F on file with the US Securities and Exchange Commission. Should one or more of these risks or uncertainties

materialize, or should underlying assumptions prove incorrect, actual results may vary materially from those anticipated, believed, estimated or expected. Novartis is providing the information in this press release as of this date and does not undertake any obligation to update any forward-looking statements contained in this press release as a result of new information, future events or otherwise.

About Novartis

Novartis provides innovative healthcare solutions that address the evolving needs of patients and societies. Headquartered in Basel, Switzerland, Novartis offers a diversified portfolio to best meet these needs: innovative medicines, eye care, cost-saving generic pharmaceuticals, preventive vaccines and diagnostic tools, over-the-counter and animal health products. Novartis is the only global company with leading positions in these areas. In 2011, the Group s continuing operations achieved net sales of USD 58.6 billion, while approximately USD 9.6 billion (USD 9.2 billion excluding impairment and amortization charges) was invested in R&D throughout the Group. Novartis Group companies employ approximately 124,000 full-time-equivalent associates and operate in more than 140 countries around the world. For more information, please visit http://www.novartis.com.

Novartis is on Twitter. Sign up to follow @Novartis at http://twitter.com/novartis.

References

- (1) The CATT Research Group, Ranibizumab and Bevacizumab for Treatment of Neovascular Age-Related Macular Degeneration Two-Year Results Ophthalmology 2012;xx:xxx. http://www.ophsource.org/webfiles/images/journals/ophtha/announcement.pdf
- (2) The CATT Research Group. Ranibizumab and Bevacizumab for Neovascular Age-Related Macular Degeneration, *New England Journal of Medicine*, April 28, 2011 (10.1056/NEJMoa1102673)
- (3) Curtis LH, Hammill BG, Schulman KA, Cousins SW. Risks of mortality, myocardial infarction, bleeding, and stroke associated with therapies for age-related macular degeneration. Arch Ophthalmol, 2010;128:1273-9.
- (4) Gower EW, Cassard S, Chu L, Varma R, Klein R. Risks of mortality, myocardial infarction, bleeding, and stroke associated with therapies for age-related macular degeneration. ARVO 2011,

- (5) http://www.mhra.gov.uk/Safetyinformation/Safetywarningsalertsandrecalls/DrugAlerts/Company-ledrecalls/CON146660
- (6) http://www.nytimes.com/2011/08/31/health/31drug.htmlPublication
- (7) http://www.fda.gov/Drugs/DrugSafety/ucm270296.htm
- (8) http://www.lakemedelsverket.se/english/All-news/NYHETER-2012/Position-of-the-Medical-Products-Agency-Regarding-/
- (9) http://webprod3.hc-sc.gc.ca/dpd-bdpp/start-debuter.do?lang=eng

Novartis Media Relations

Central media line : +41 61 324 2200

Eric Althoff Julie Morrow

Novartis Global Media Relations Novartis Pharma Communications

+41 61 324 7999 (direct) +41 61 696 7581 (direct)

+41 79 593 4202 (mobile) +41 79 357 3259 (mobile)

eric.althoff@novartis.com julie.morrow@novartis.com

e-mail: media.relations@novartis.com

For Novartis multimedia content, please visit www.thenewsmarket.com/Novartis For questions about the site or required registration, please contact: journalisthelp@thenewsmarket.com.

Novartis Investor Relations

Central phone:	+41 61 324 7944		
Susanne Schaffert	+41 61 324 7944	North America:	
Pierre-Michel Bringer	+41 61 324 1065	Helen Boudreau	+1 212 830 2404
Thomas Hungerbuehler	+41 61 324 8425	Jill Pozarek	+1 212 830 2445
Isabella Zinck	+41 61 324 7188	Edwin Valeriano	+1 212 830 2456

e-mail: investor.relations@novartis.com e-mail: investor.relations@novartis.com

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Novartis AG

Date: May 1, 2012 By: /s/ MALCOLM B. CHEETHAM

Name: Malcolm B. Cheetham Title: Head Group Financial

Reporting and Accounting

6